Author's response to reviews

Title: Macrophage traits in cancer cells are induced by macrophage-cancer cell fusion and cannot be explained by cellular interaction

Authors:

Ivan Shabo (ivan.shabo@liu.se)
Kristine Midtbø (krimi000@student.liu.se)
Henrik Andersson (Henrik.A.Andersson@lio.se)
Emma Åkerlund (emma.akerlund92@gmail.com)
Hans Olsson (hans.olsson@lio.se)
Pia Wegman (pia.wegman@lio.se)
Cecilia Gunnarsson (cecilia.gunnarsson@lio.se)
Annelie Lindström (annelie.lindstrom@liu.se)

Version: 2
Date: 14 December 2014

Author's response to reviews: see over
Dear Madame/Sir,

I changed the text of the manuscript in accordance with your requirements as below:

- I added line numbers throughout the document.
- I included the email addresses for all authors in tittelsidan of the manuscript.
- I note in the method text of the study is approved by the ethics committee with reference numbers for their approval.

With best regards

Ivan Shabo  MD PhD
Inst of Clinical and Experimental Medicine
Linköping University
Sweden